\documentclass[12pt]{article}
\usepackage{fullpage}
\usepackage{hyperref}
\renewcommand{\familydefault}{\sfdefault}
\addtolength{\topmargin}{-0.5in}
\addtolength{\textheight}{1in}
\addtolength{\oddsidemargin}{-0.5in}
\addtolength{\evensidemargin}{-0.5in}
\addtolength{\textwidth}{1in}



\begin{document}

\begin{center}
{\Large \textbf{Applied Mechanism Design for Social Good}}\\
CMSC828M -- Fall 2016 -- University of Maryland\\
Scribe: Your Name Here
\end{center}

\section*{Overview of Lectures 6 and 7: Security Games}
Write a parapgraph or two that, at a high level, describes the topic of the lecture(s).  

\section*{Details}

Give a nice play-by-play of the paper(s), lecture(s), and any discourse from the class.  Include whatever you think is necessary for the reader of these notes to take away from reading this document a good understanding of the subject matter we covered.

\end{document}
